Author's response to reviews

Title: The respiratory health of urban Indigenous children aged less than 5 years: study protocol for a prospective cohort study

Authors:

Kerry K Hall (kk.hall@qut.edu.au)
Anne B Chang (anne.chang@menzies.edu.au)
Theo P Sloots (t.sloots1@uq.edu.au)
Jennie Anderson (jennie@murrimedical.com.au)
Anita Kemp (anita@murrihealthgroup.org.au)
Janet Hammill (janet.hammill@uq.edu.au)
Michael Otim (michael.otim@acu.edu.au)
Kerry-Ann F O'Grady (kerryann.ogrady@qut.edu.au)

Version: 3 Date: 1 March 2015

Author's response to reviews:

1. Authors emails have been added as requested

2. Author Contribution have been added as requested